Author's response to reviews

Title: Rate of first recorded diagnosis of autism and other pervasive developmental disorders in United Kingdom general practice, 1988 to 2001

Authors:

Dr Liam Smeeth (liam.smeeth@lshtm.ac.uk)
Ms Claire Cook (claire.cook@lshtm.ac.uk)
Eric Fombonne (eric.fombonne@staff.mcgill.ca)
Lisa Heavey (nunsnunsreverse@aol.com)
Laura Rodrigues (laura.rodrigues@lshtm.ac.uk)
Peter Smith (peter.smith@lshtm.ac.uk)
Prof Andy Hall (andy.hall@lshtm.ac.uk)

Version: 2 Date: 18 Feb 2004

not applicable